Author's response to reviews

Title: Rationale and methods of the European Study on Cardiovascular Risk Prevention and Management in Daily Practice (EURIKA)

Authors:

Fernando Rodríguez Artalejo (fernando.artalejo@uam.es)
Eliseo Guallar (eguallar@jhsph.edu)
Claudio Borghi (claudio.borghi@unibo.it)
Jean Dallongeville (jean.dallongeville@pasteur-lille.fr)
Guy De Backer (guy.debacker@ugent.be)
Julian P Halcox (halcoxjp@cardiff.ac.uk)
Ramón Hernández-Vecino (Ramon.Hernandez@astrazeneca.com)
Francisco Javier Jiménez (javier.jimenez@astrazeneca.com)
Elvira L Massó-González (Elvira.MassoGonzalez@astrazeneca.com)
Joep Perk (joep.perk@lnu.se)
Philippe Gabriel Steg (gabriel.steg@bch.aphp.fr)
José R Banegas (joseramon.banegas@uam.es)
EURIKA investigators EURIKA investigators (fernando.artalejo@uam.es)

Version: 5 Date: 27 June 2010

Author’s response to reviews:

Description of formatting changes made in the manuscript BMC Public Health, MS: 5110907733766519

MAJOR REVISIONS

In the authors list, we have replaced the term “on behalf of EURIKA investigators” has been replaced by “for the EURIKA investigators”. Moreover, we have included an additional author ("EURIKA investigators") in the authors list within the submission system.

MINOR REVISIONS

Figure 1 has been cropped as closely as possible to minimize white space around the image.

All additional files have been removed.

Thank you so much for your editorial support